Author's response to reviews

Title: An electronic clinical trial tool to recruit large patient samples and assess selection bias in general practice research

Authors:

Stephanie Heinemann (sheinem3@gwdg.de)
Sabine Thüring (sthueri@gwdg.de)
Sven Wedeken (swedeken@gwdg.de)
Tobias Schäfer (tobias.schaefer@medizin.uni-goettingen.de)
Christa Scheidt-Nave (Scheidt-Nave@rki.de)
Mirko Ketterer (mirko.ketterer@it-choice.de)
Wolfgang Himmel (whimmel@gwdg.de)

Version: 2 Date: 16 April 2010

Author's response to reviews:

We have submitted the revised version in MS Word's correction mode. We can of course also submit a "clean" revised version. We have chosen not to do so, so that there would be no confusion about the page and line numbers to which we refer in our authors' comment document. Please let us know what you would prefer.